Author’s response to reviews

Title: Timing of first antenatal care attendance and associated factors among pregnant women in public health institutions of Axum Town, Tigray, Ethiopia, 2017: A mixed design study

Authors:
Berhanu Gebrehiwot (brhangs19@gmail.com)
Tilahun Belete (tilahunbe100@gmail.com)
Weyzer Tilahun (weyzer.tilahun@gmail.com)
Betell Tafere (betiela03@gmail.com)
Senait Gebremeskel (senaitgs42@gmail.com)

Version: 1 Date: 25 Sep 2018

Author’s response to reviews:

Response to reviewers

Title: Timing of first Ante Natal Care Attendance and associated Factors among pregnant Women in Public Health institutions of Axum Town, Tigray, Ethiopia, 2017: A Cross sectional study with qualitative and quantitative approaches

Dear reviewers,

Thank you for your constructive comments and suggestions. We found your comments very helpful for the improvement of our manuscript and learn about some of the limitations of our study. We have tried to address the comments raised and incorporated in the document.